Pyrethroid-Resistance and Presence of Two Knockdown Resistance (*kdr*) Mutations, F1534C and a Novel Mutation T1520I, in Indian *Aedes aegypti*



Raja Babu S. Kushwah¹, Cherry L. Dykes¹, Neera Kapoor², Tridibes Adak¹, Om P. Singh¹*

1 National Institute of Malaria Research, Sector 8, Dwarka, Delhi, India, 2 School of Life Sciences, Indira Gandhi National Open University, Maidangarhi, New Delhi, India

Abstract

Background: Control of Aedes aegypti, the mosquito vector of dengue, chikungunya and yellow fever, is a challenging task. Pyrethroid insecticides have emerged as a preferred choice for vector control but are threatened by the emergence of resistance. The present study reports a focus of pyrethroid resistance and presence of two kdr mutations—F1534C and a novel mutation T1520I, in Ae. aegypti from Delhi, India.

Methodology/Principal Findings: Insecticide susceptibility status of adult-female *Ae. aegypti* against DDT (4%), deltamethrin (0.05%) and permethrin (0.75%) was determined using WHO's standard insecticide susceptibility kit, which revealed resistance to DDT, deltamethrin and permethrin with corrected mortalities of 35%, 72% and 76% respectively. Mosquitoes were screened for the presence of *kdr* mutations including those reported earlier (I1011V/M, V1016G/I, F1534C, D1794Y and S989P), which revealed the presence of F1534C and a novel mutation T1520I. Highly specific PCR-RFLP assays were developed for genotyping of these two mutations. Genotyping using allele specific PCR and new PCR-RFLP assays revealed a high frequency of F1534C (0.41–0.79) and low frequency of novel mutation T1520I (0.13). The latter was observed to be tightly linked with F1534C and possibly serve as a compensatory mutation. A positive association of F1534C mutation with DDT and deltamethrin resistance in *Ae. aegypti* was established. However, F1534C-*kdr* did not show significant protection against permethrin.

Conclusions/Significance: The Aedes aegypti population of Delhi is resistant to DDT, deltamethrin and permethrin. Two kdr mutations, F1534C and a novel mutation T1520I, were identified in this population. This is the first report of kdr mutations being present in the Indian Ae. aegypti population. Highly specific PCR-RFLP assays were developed for discrimination of alleles at both kdr loci. A positive association of F1534C mutation with DDT and deltamethrin resistance was confirmed.

Citation: Kushwah RBS, Dykes CL, Kapoor N, Adak T, Singh OP (2015) Pyrethroid-Resistance and Presence of Two Knockdown Resistance (kdr) Mutations, F1534C and a Novel Mutation T1520I, in Indian Aedes aegypti. PLoS Negl Trop Dis 9(1): e3332. doi:10.1371/journal.pntd.0003332

Editor: Pattamaporn Kittayapong, Mahidol University, Thailand

Received June 3, 2014; Accepted October 11, 2014; Published January 8, 2015

Copyright: © 2015 Kushwah et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability: The authors confirm that all data underlying the findings are fully available without restriction. All relevant data are within the paper, and are available at GenBank accession numbers KM677247–KM677334.

Funding: RBSK was supported by Indian Council of Medical Research (ICMR), Senior Research Fellowship grant F/810/2010-ECD-II and CLD was supported by National Institutes of Health grant U19AI089676. Partial financial support was provided by Department of Biotechnology (Government of India). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* singh@mrcindia.org

Introduction

Aedes aegypti is globally distributed throughout the tropics and subtropics and highly adapted to humans and urban environments. It acts as a primary vector for various arboviral infections including yellow fever virus, dengue virus (DENV) and chikungunya virus (CHIKV) [1]–[5]. Dengue has recently become a major health problem around the world with more than 120 countries endemic for dengue [6] and has been ranked as the most important mosquito borne viral disease [7]. Recent estimates by the World Health Organization (WHO) suggests that 50–100 million dengue infections occur worldwide every year and over 40% of the world's population is now at risk of the disease [8]. A study based on a cartographic approach estimated 90 million apparent dengue infections globally in year 2010 with India accounting for 34% (32 million) infections [9]. Chikungunya is another important arboviral infection spread by *Ae. aegypti*, prevalent in Africa, Southeast Asia and India [10]. In India it reemerged in 2006 after a gap of 32 years [10].

Since there is no specific vaccine or drug available for the treatment of dengue and chikungunya, vector control and personal protection are the only options to reduce the spread of these arboviral infections. Vector control strategies employed for *Aedes* control in India are mainly anti-larval measures, source reduction and use of adulticides (pyrethrum space spray and malathion-fogging) during a disease outbreak. Pyrethroids are widely used for personal protection in the form of repellents and insecticide treated materials [11], which provides effective protection against day biting *Aedes*. It has also been shown that window curtains and domestic water container covers treated with insecticide may reduce densities of dengue vectors to low levels and potentially affect dengue transmission [12]. In addition pyrethroids have been

Author Summary

Dengue and chikungunya are the two important human arboviral infections in India transmitted mainly by Aedes aegypti. In absence of any specific drug or vaccine for these infections, vector control and personal protection are the only control options available. The success of insecticide-based vector control heavily relies upon the knowledge of the status of insecticide resistance in vector populations and the underlying mechanisms of insecticide resistance, especially in the presence of cross-resistance. Knockdown resistance (kdr) is one of the mechanisms of resistance that confers cross-resistance to DDT and pyrethroids. Currently, pyrethroids are the only insecticide class recommended for use in long-lasting insecticide nets (LLIN) and have proven superior to all other insecticides used in vector control programme, due to low mammalian toxicity, low residual activity in nature and rapid knockdown action. The present study was undertaken to determine the susceptibility status of Ae. aegypti against DDT and pyrethroids, and identification of kdr mutations. Though the presence of kdr mutations in Ae. aegypti has been reported in many countries, such a report is not available from India. This study for the first time reports the presence of two kdr mutations, F1534C and a novel mutation T1520I, in an Indian Ae. aegypti population.

recommended by WHO for space spraying for *Aedes* control [13] due to rapid knockdown effect and less mammalian toxicity. However, the use of pyrethroids is being challenged by the rapid emergence of resistance, which needs to be monitored periodically to manage effective programmes to avoid or delay resistance in vector species. Key to this is, understanding of the mechanisms of resistance so that informed decisions can be made to select appropriate insecticides for effective control of target vector species.

One of the mechanisms of resistance in insects against DDT and pyrethroids is knockdown resistance (kdr) which is conferred by mutation(s) in the target site, the voltage gated sodium channel (VGSC). Several kdr mutations have been reported in many insects of agricultural and medical importance including Ae. aegypti. In Ae. Aegypti, eleven non-synonymous mutations at nine different loci have been reported [14]-[17], amongst which mutations at three loci, i.e., Iso1011 (I \rightarrow M/V) and Val1016 $(V \rightarrow G/I)$ in domain II and F1534 $(F \rightarrow C)$ in domain III are most commonly reported as contributing to pyrethroid resistance [14]-[22]. The most common kdr-mutations L1014F/S reported in many insects of agricultural and medical importance is not yet found in Ae. aegypti possibly due to codon constraint [23]. Although widespread in Southeast Asia and Latin America, the presence of kdr mutations has yet to be established in India. Here we report the presence of two kdr mutations, F1534C and a novel mutation T1520I, in an Indian Ae. aegypti population.

Materials and Methods

Mosquito collection

Aedes immature (larvae and pupae) were collected from the water holding containers in domestic and peri-domestic areas in Delhi and were reared to adults. The collection sites and dates of collections are shown in supplemental items S1 Table and S1 Fig. Ground mixtures of dog biscuits and fish food in a ratio of 3:1 were provided as food for larvae. Emerged adults were identified morphologically and supplied with 10% glucose solution soaked in cotton pads.

Insecticide susceptibility bioassay

Two-to four-days old adult *Ae. aegypti* female mosquitoes were subjected to insecticide susceptibility testing using the WHO's standard insecticide susceptibility test kit. Up to twenty-five mosquitoes in each replicate were exposed to 4% DDT, 0.05% deltamethrin and 0.75% permethrin impregnated paper (supplied by WHO collaborative centre, Vector Control Research, Universiti Sains, Malaysia) alongside appropriate controls for one hour and subsequently transferred to recovery tubes lined with untreated paper. During recovery, mosquitoes were provided access to cotton soaked in 10% glucose and mortalities were recorded after 24 hours. All the bioassays were carried out at $27\pm1^{\circ}$ C and $70\pm10\%$ relative humidity. Percent mortalities were calculated using Abbott's formula [24]. Dead and alive mosquitoes after recovery was transferred to individual microfuge tubes and stored at -20° C.

DNA isolation and kdr genotyping

DNA was isolated from individual mosquitoes following Livak *et al.* (1984) [25]. Allele specific PCR assays were employed for genotyping of *kdr* mutations I1011V/M, V1016G/I and F1534C following Saavedra *et al.* (2007) [16] and Yanola *et al.* (2011) [26]. For genotyping of D1794Y, PCR-RFLP was carried out as described by Chang *et al.* (2012) [27]. In the absence of established PCR-based assays for mutation S989P, direct sequencing was carried out using primers IIP_F and IIS6_R [26]. Dead as well as surviving mosquitoes of some batches of insecticide bioassay tests were genotyped for F1534C mutation to study the association of this *kdr* mutation with insecticide resistance.

DNA sequencing

DNA sequencing was performed to validate the PCR-based genotyping used for various kdr alleles and also to check for the presence of any novel mutation. Three regions of VGSC were amplified and sequenced: (i) partial domain II (P to S6) using primers IIP_F and IIS6_R [26], (ii) partial domain III (S4-S6) using primers Ge-IIIS6_F and IIIS6R [26], and (iii) partial domain IV (S5-S6) using primers 5380F1 and 5380R1 [27]. PCR products were amplified, purified using QIaquick PCR purification kit (Qiagen Inc) and subjected to cycle sequencing reaction using BigDye Terminator v3.0. The termination products were run in Applied Biosystems 3730×l DNA Analyzer. Some of the sequencing reactions were performed at Macrogen Inc (South Korea). Sequencing chromatograms were edited using FinchTV ver 1.5.0 (Geospiza, Inc.). The PCR product of one sample, which was suspected to have an indel in the intron, was cloned in pGEM-T vector system using vendor's protocol and five clones were sequenced. Sequences were aligned using ClustalW implemented in Mega5 [28].

Development of PCR-RFLP for kdr genotyping

For development of PCR-RFLP assays for detection of *kdr* alleles at two loci (F1534 and T1520) in domain III-S6, DNA sequences spanning 200 bp upstream to F1534 and 200 bp downstream to T1520 were checked for 1534C- and 1520I-specific restriction sites using an online tool available at http://insilico.ehu.es/restriction/two_seq. Two unique restriction enzymes *SsiI* and *BsaBI* were selected which were specific to1534C (TTC>TGC) and 1520I (ACC>ATC) sequences respectively. The intron region was excluded when designing PCR-RFLP due to the existence of indel in the intron upstream of T1520 as revealed by sequencing of cloned PCR product. Two primers

flanking these two loci, i.e., AekdrF (5'-TGGGAAAGCAGCC-GATTC-3') and AekdrR (5'-CCTCCGTCATGAACATTTCC-3') were designed with expected amplicon size of 171 bp. The expected sizes of cleaved product for the 1520I allele were 143 and 28 bp when digested with *BsaB*I, and 103 and 68 bp for 1534C when digested with *SsiI*. The diagnostic criterion for 1520I allele was taken as the presence of 143 bp band only (resolution of 28 bp cleaved product can not be resolved on agarose gel), whereas presence of 103 and 68 bp bands were considered as diagnostic criteria for 1534C allele. Uncut product of 171 bp was considered the wild allele.

For PCR-RFLP, amplification was carried out in 15 μ l of reaction mixture containing 1× buffer, 200 μ M of each dNTP, 0.25 μ M of primers AekdrF and AekdrR and 0.5 unit of *Taq* DNA polymerase. The PCR conditions were initial denaturation at 95°C for 3 min followed by 35 cycles each of denaturation at 95°C for 15 s and extension at 72°C for 30 s and a final extension at 72°C for 7 min. The PCR product was subjected to two separate restriction digestion reactions, one with *BsaBI* and another with *SSiI*. Each restriction digestion reaction mixture (20 μ l) contained 5 μ l of PCR product, 2 units of restriction enzyme and 1× buffer, which was incubated for four hours or overnight at 65°C for *BsaBI* and 37°C for *SSiI*. The cleaved product was run on 2.5% agarose gel containing ethidium bromide and visualized with a gel documentation system (Figs. 1 and 2).

Representative samples of PCR-RFLP genotyped samples were sequenced for partial domain III to validate PCR-RFLP result where primers AekdrF and AekdrR were used for amplification of PCR product and only AekdrR was used for sequencing reaction.

DNA sequences with read length of 200 bp or more have been deposited in GenBank (accession numbers: KM677247– KM677334).

Statistical analysis

The association of the *kdr* mutations with resistance phenotype was tested using Fishers' Exact test and Odds Ratio estimation using dominant, recessive and additive models. Hardy-Weinberg equilibrium test was performed using Chi-square or Fisher's Exact

test. Analysis of linkage disequilibrium and the Hill and Robertson coefficient r^2 were calculated for alleles using CubeX software (http://www.oege.org/software/cubex) [29].

Results

Insecticide susceptibility status

The result of insecticide susceptibility tests carried out on Ae. *aegypti* against DDT, deltamethrin and permethrin are shown in Table 1. The result shows high resistance against DDT (30.2–48.1% mortality) and moderate level of resistance to pyrethroids (deltamethrin: 64.4–74.3%; permethrin: 66.8–82.3% mortalities) in all sites.

Genotyping of kdr alleles

Results of allele specific PCR genotyping for the F1534C mutation are shown in Table 1. The allelic frequency of the 1534C mutant is high in all the three sites ranging from 41—69%. Of the1180 samples genotyped, a total of 34 sample representing FF (n = 9), FC (n = 11) and CC (n = 14) were sequenced for partial domain III to validate allele specific PCR results. Two samples showed discrepancies where homozygous CC turned out to be FC after sequencing. Sequencing of samples also revealed the presence of a novel mutation C>T on the second codon of T1520 residue (ACC) leading to T→I amino acid substitution. Among 34 samples sequenced for partial domain III, eleven were with FC/TT, two with FC/TI, eleven with CC/TT, one with CC/II and the remaining nine were with FF/TT.

A total of 166 samples were genotyped for I1011M/V and V1016G/I. Though some samples were observed to be positive for mutations (genotypes IM = 7, MM = 1, IV = 2 for I1011 locus, and VI = 6 for V1016 locus) by allele specific PCR, sequencing of 29 samples representing all genotypes (all mutants and 13 wild genotypes) did not confirm the presence of any of them. Sequencing of partial domain II also did not identify S989P-*kdr* mutation in any sample. The genotyping for I1011 and V1016 was therefore discontinued assuming that allele specific PCR is not specific and I1011M/V or V1016G/I mutations are absent in the study population. For genotyping of D1794Y, a total of 66



Fig. 1. Gel photograph showing PCR-RFLP assay for genotyping of T1520 alleles. Lane 1: 100 bp DNA ladder, lanes 2–3: TT, lanes 4–5: TI heterozygotes, lanes 6–7: II, lane 8: negative control. doi:10.1371/journal.pntd.0003332.q001



Fig. 2. Gel photograph showing PCR-RFLP assay for genotyping of F1534 alleles. Lanes 1 and 9: 100 bp DNA ladder, lanes 2–3: FF, lanes 4– 5: FC heterozygotes, lanes 6–7: CC, lane 8: negative control.

doi:10.1371/journal.pntd.0003332.g002

mosquitoes were genotyped using PCR-RFLP and five samples through DNA sequencing, but all turned out to be the reference genotype.

Association of F1534C mutation with insecticide resistance

The distribution of different F1534 is shown in Table 2. The proportions of dead and live mosquitoes after exposure to insecticides for each genotype are shown in Fig. 3. Odds Ratio (OR) estimates at 95% confidential intervals (CI) and Fisher's exact test using different models (dominant, recessive and additive) for dead and live mosquitoes in each treatment group are presented in Table 3. It was observed that F1534C-*kdr* conferred greater protection against DDT with all models and highest protection was shown using the recessive model (OR = 16.0, 95% CI: 5.6–45.4; p = 0.000). Lower protection was shown against deltamethrin when fitted with recessive (OR = 2.0, 95% CI: 1.06–3.75; p < 0.05) or additive (OR = 1.85, 95% CI: 1.84–2.89; p < 0.01) models. However, F1534C-*kdr* did not show significant protection against permethrin.

Genotyping of F1534 and T1520 alleles using new PCR-RFLP

Genotyping of F1534 and T1520 alleles were performed on 203 mosquitoes, which revealed a high frequency of the F1534C mutation (0.79) and a very low frequency of the T1520I mutation (0.13). Genotyping results showing association of T1520 and F1534 alleles are shown in Table 3. It was observed that T1520I mutation was found in individuals having the 1534C allele only, but never with wild type F1534. This data infers that 1520I is linked to 1534C. Linkage disequilibrium (LD) analysis revealed perfect disequilibrium (D' = 1.0, χ^2 = 8.02) though r² was low (0.04) due to a relatively low frequency of allele 1520I as compared to 1534C, where all individuals with 1520I allele showed association with 1534C, but not all 1534C are associated with 1520I. The present data revealed the presence of three haplotypes with haplotype frequencies $f_{\rm TF}$ = 0.21, $f_{\rm TC}$ = 0.66 and $f_{\rm IC}$ = 0.13. However, $f_{\rm IF}$ was absent.

Validation of new PCR-RFLP assays

Among the samples genotyped using the new PCR-RFLP, a portion of domain III was sequenced for 20 samples (two sample of TT/CC, five samples of TI/FC, eleven samples of TI/CC and two samples of II/CC). Genotyping results agreed with DNA sequencing results.

Discussion

Vector control is the only option for suppression of Ae. aegyptiborne dengue and chikungunya infections in the absence of vaccine or drugs. Several pyrethroids have been recommended by WHO for use in space spray against Aedes [13]. Though in India pyrethroids are being extensively used in malaria control programme, their use in urban areas is limited to space spraving of pyrethrum and fogging with malathion. However, pyrethroidbased household anti-mosquito gadgets (liquid vaporizer, mats, coils) are extensively used as personal protectants against mosquito nuisance. Use of these devices may be contributing to resistance in Ae. aegypti in Delhi. Insecticide susceptibility tests carried out in India by several authors prior to the year 2014 did not reveal pyrethroid resistance, though they were found to be resistant to DDT [30]-[33]. Only in one case, 2% survival was recorded in Aedes aegypti on exposure to diagnostic concentration of deltamethrin in a strain from Jharkhand, India [30]. Very recently, for the first time in India, resistance to pyrethroids has been reported from Assam state [34].

In the absence of baseline insecticide susceptibility data for Indian *Ae. aegypti* or universally acceptable discriminating dose for *Ae. aegypti*, we used 4% DDT, 0.05% deltamethrin and 0.75% permethrin papers for bioassays, the most frequently cited doses in recent publications [30]–[31], [34]–[40] to facilitate easy comparison. Previous published data from Delhi showed that *Ae. aegypti* was 100% susceptible to even lower doses of insecticides, i.e., 0.025% deltamethrin and 0.25% permethrin [31]. Less than 80% mortalities of mosquitoes at higher doses (which are 2-fold and 3-fold respectively) confirm resistance against these insecticides.

The extensive use of insecticides for vector control has raised concern over the development of insecticide resistance and adverse Table 1. Result of insecticide susceptibility test against DDT, deltamethrin (DEL) and permethrin (PER), and genotyping result of F1534 alleles as determined by allele specific PCR.

Locality	Percent corrected m	ortality (replicates/n)		F1534 ger	otypes			Allelic frequenc	ies	<i>р</i> * (НWE)
	DDT 4%	DEL 0.05%	PER 0.75%	۲.	ñ	ម	Total	F1534	1534C	
South Delhi I	30.17% (20/348)	71.86% (12/231)	82.31% (19/373)	118	195	214	527	0.409	0.591	0.000
South Delhi II	37.58% (9/165)	74.32% (8/148)	66.79% (14/265)	35	128	158	321	0.308	0.692	0.504
West Delhi	48.15% (6/108)	64.41% (3/59)	74.74% (5/95)	139	112	81	332	0.587	0.413	0.000
Pooled data	35.27% (35/621)	71.69% (23/438)	75.72% (38/733)	292	435	453	1180	0.432	0.568	0.000
HWE = Hardy-Weinberg equ	ilibrium									

https://www.competible.com/ *chi-square test. doi:10.1371/journal.pntd.0003332.t001 Table 2. Association of F1534 alleles with insecticide resistance phenotypes.

Insecticides		Genotyp	e		Odds ratio (95% Cl)			Fisher's exact test (µ	y value)	
		ŧ	Я	y	Recessive model	Dominant model	Additive model	Recessive model	Dominant model	Additive model
DDT	Dead	38	26	4	16 (5.64–45.42)	5.72 (3.18–10.30)	5.81 (3.76–8.96)	< 0.0001	< 0.0001	< 0.0001
	Alive	41	72	113						
Deltamethrin	Dead	51	55	46	2.0 (1.06–3.75)	2.1 (0.99–4.33)	1.85 (1.84–2.89)	< 0.05	NS	<0.01
	Alive	11	19	26						
Permethrin	Dead	50	59	66	1.05 (0.58–1.89)	0.77 (0.37–1.61)	1.37 (0.89–2.14)	NS	NS	NS
	Alive	6	18	31						
NS = non-significant. doi:10.1371/journal.pntd.00	J03332.t002									



Fig. 3. Proportion of dead and alive mosquitoes in each genotype for F1534 alleles exposed to DDT 4%, deltamethrin 0.05% and 0.75% permethrin for one hour. doi:10.1371/journal.pntd.0003332.g003

effects on the environment and human health [41]. Genes conferring insecticide resistance have been spreading in vector populations, particularly in vectors of pathogens causing malaria and dengue [42]. The fact that dispersal of Aedes may be more rapid than other mosquitoes due to transportability of dried, but viable eggs through containers, a single resistance mechanism can spread rapidly. Knockdown resistance (kdr) is one of the mechanisms of DDT and pyrethroid resistance in insects. It is conferred by amino acid substitution(s) in the target site (VGSC) resulting in reduced sensitivity of the target site. A number of mutations have been reported in the VGSC of Ae. aegypti across Latin America and Southeast Asia amongst which V1016G/I, I1011M/V and F1534C [14], [18], [21], [22], [15], [43] are known to confer resistance. F1534C has been reported from Latin America [44], [22], [45] and Southeast Asia [15], [43], [46], and shown to confer resistance against DDT and pyrethroids. In our study we provide evidence that this mutation confers a high level of protection against DDT and relatively low protection against deltamethrin. However, we failed to show significant protection against permethrin. Our failure to establish association of F1534C with permethrin resistance is contrary to findings by Harris et al. (2010) [22] and Yanola et al., (2011) [15]. Our result is also contradictory to the findings of Du et al., (2013) [47] who were able to demonstrate that F1534C reduced the channel sensitivity to permethrin but not against deltamethrin when expressed in Xenopus oocyte. Failure to establish association of F1534C with permethrin resistance is surprising and needs to be further investigated. It may be possible that the dose of permethrin (0.75%) used for discrimination of kdr-resistant mosquito in the Indian population is too high that might have killed kdr-resistant mosquitoes. Another possible reason for such discrepancy may be due to the presence of some other linked mutations. For example, F1534C has shown to be strongly associated with permethrin in Grand Cayman [22] where another mutation V1016I co-existed. It is possible that protection against permethrin in Grand Cayman may be due to the combined effect of F1534C and V1016I. It is interesting to explore such association because such linkage has been shown in Brazil, where V1016I was always associated with F1534C [44]. To know the exact role of any particular kdr mutation, one should perform such association studies using laboratory lines of mosquitoes characterized for complete VGSC sequence.

In this study we explored a novel mutation T1520I in an Indian *Ae. aegypti* population. Its potential role in resistance to insecticides is yet to be ascertained. However, since this mutation has always been found in association with F1534C mutation (D' = 1), it may be a compensatory mutation to reduce the fitness cost from possible deleterious effects of F1534C mutation, though it has been shown through laboratory experiments that *Ae. aegypti* homozygous for F1534C does not have reduced fitness [48]. However, its additive effect on protection against DDT and pyrethroids cannot be ruled out. We also observed that 1520I is

Table 3. Genotyping results of PCR-RFLP assays for F1534C and T1520I alleles and their association.

		F1534 ge	enotypes			
		FF	FC	сс	Total	
T1520 genotypes	TT	28	22	105	155	
	TI	0	8	37	45	
	II	0	0	3	3	
	Total	28	30	145	203	

pHWE (Fisher's exact test): T1520 alleles = 0.991; F1534 alleles = 0.000. doi:10.1371/journal.pntd.0003332.t003

associated with 1534C, but not vice versa explaining the low r^2 (0. 039), which reflects a very low frequency of T1520I as compared to F1534C. Whether haplotype 1534C/1520I is under positive selection remains to be established. Interestingly a similar linkage association is found in the Brazilian population where 1016I is associated with 1534C, but not vice versa [44]. These different associations in different geographical locations indicate that the most likely association of T1520I in India and V1016I in Brazil with F1534C are under positive selection. Linss et al. 2014 [44], noted a progressive increase of the Na_V^{R2} haplotype (double mutant, F1534C with V1016I) from year 2002 through 2012 and concluded that it is likely to be the most favourably selected allele. Linkage of kdr mutations is very common in Ae. aegypti. Cosegregation has been shown between 1016G with D1794Y [17], 1016G with 989P [49] and 1016G with 1534C [22], [46]. Whether positive selection of such linkage associations is due to additive role in protection against insecticides or due to compensatory advantage, is worth investigating. Since novel mutation T1520I is tightly linked to F1534C and homozygotes are found in very low frequency, the exact role of this novel mutation could not be established.

Our result shows that F1534 genotypes show significant deviation from Hardy-Weinberg equilibrium in all populations (p < 0.0001) except in South Delhi-II. Initially we thought that this might be due to discrepancy in allele specific PCR genotyping, which often fails to prevent non-specific annealing during PCR extension. However, when we carried out genotyping using highly specific PCR-RFLP method, there was no change in HWE parameter for F1534 alleles. Surprisingly, T1520 genotypes in the same group of mosquitoes were in perfect HWE (p = 0.99). The possible explanation for such a deviation may be the presence of heterogeneous populations or gene duplication. Further studies are required to resolve this issue.

Knockdown resistance, which is known to confer crossresistance to DDT and pyrethroids cannot be monitored through routine insecticide susceptibility tests and requires a sensitive and reliable molecular method of detection. PCR-based methods, allele specific PCR in particular, are most widely used method for this purpose. The specificity of allele specific PCRs which are based on single nucleotide polymorphism is often compromised due to the fact that single base mismatch often does not prevent extension [50] and leads to non-specific amplification [26,51]. Yanola *et al.*, (2011) reported an overestimation of 1534C

References

- Reed W (1901) Propagation of yellow fever: observation based on recent researches. Medical Record 60: 201–209.
- Bancroft TL (1906) On the aetiology of dengue fever. Australian Medical Gazette 25: 17–18.
- Cleland JB, Bradley B, McDonald W (1918) Dengue fever in Australia. Its history and clinical course, its experimental transmission by *Stegomyia fasciata*, and the results of inoculation and other experiments. J Hyg 16: 317–418.
- Ross RW (1956) The Newala epidemic III; the virus: isolation, pathogenic properties and relationship to the epidemic. J Hyg 54: 177–191.
- Pialoux G, Gauzere BA, Jaureguiberry S, Strobel M (2007) Chikungunya, an epidemic arbovirosis. Lancet Infect Dis 7: 319–327.
- Ng LC (2011) Challenges in dengue surveillance and control. Western Pac Surveill Response J 2: 1–3.
- WHO (2012) Global strategy for dengue prevention and control 2012–2020. Geneva, Switzerland.
- WHO (2014) Dengue and severe dengue. Fact sheet N°117 (Updated March 2014) (website: http://www.who.int/mediacentre/factsheets/fs117/en/, Retrieved: 25 May 2014).
- Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, et al. (2013) The global distribution and burden of dengue. Nature 496: 504–507.
- Lahariya C, Pradhan SK (2006) Emergence of chikungunya virus in Indian subcontinent after 32 years: A review. J Vector Borne Dis 43: 151–160.
- Rozendaal JA (1997) Vector control: Methods for use by individuals and communities. World Health Organization, Geneva.

frequency by 1.8% while using allele specific PCR [26]. In the present study, non-specific amplification was evident in the allele specific PCRs we employed for identification of various *kdr* alleles located at different loci. We therefore opted to develop a PCR-RFLP method for the identification of F1534 and T1520 alleles, which is presumed to be specific owing to the fact that restriction enzymes are highly specific. The sequencing of representative samples of PCR-RFLP genotyped samples (n = 20) showed 100% specificity of the assay. An additional advantage of the PCR-RFLP over allele specific PCR was that a single PCR amplicon could be used for detection of four alleles present at two loci since two mutations T1520I and F1534C are in proximity, whereas for allele specific PCR assays normally two PCR reactions are required to be performed for detecting four alleles at two loci.

The emergence of pyrethroid resistance in Indian *Ae. Aegypti* associated with the presence of the F1534C-*kdr* mutation is a threat to the success of pyrethroid-based *Aedes* control and necessitates countrywide monitoring of insecticide resistance and mapping the distribution of F1534C and T1520I mutations. Though F1534C mutation is shown to be associated with DDT and pyrethroids, the role of novel mutation T1520I still remains to be investigated.

Supporting Information

S1 Fig. Geographical locations of mosquito collection sites. (PDF)

S1 Table Geographical locations and dates of mosquito collection. (DOCX)

Acknowledgments

The authors are grateful to Dr. Sujatha Sunil and Dr. Anil Sharma for helping in sample collection, and to Mr. Uday Prakash, Mr. Shri Bhagwan, Mr. NS Bhakuni and Mrs. Sushmita Banerjee for excellent laboratory assistance. Thanks are also due to the anonymous reviewer who edited the manuscript to improve English language.

Author Contributions

Conceived and designed the experiments: OPS. Performed the experiments: RBSK CLD. Analyzed the data: OPS RBSK. Contributed reagents/materials/analysis tools: OPS. Wrote the paper: OPS NK TA.

- Kroeger A, Lenhart A, Ochoa M, Villegas E, Levy M, et al. (2006) Effective control of dengue vectors with curtains and water container covers treated with insecticide in Mexico and Venezuela: cluster randomized trials. BMJ 332: 1247–1252.
- WHO (2006) Pesticides and their Application for the Control of Vectors and Pests of Public Health Importance, 6th edn. WHO/CDS/NTD/WHOPES/ GCDPP/2006.1, Geneva.
- Brengues C, Hawkes NJ, Chandre F, McCarroll L, Duchon S, et al. (2003) Pyrethroid and DDT cross-resistance in *Aedes aegypti* is correlated with novel mutations in the voltage-gated sodium channel gene. Med Vet Ent 17: 87–94.
- Yanola J, Somboon P, Walton C, Nachaiwieng W, Prapanthadara L (2010) A novel F1552/C1552 point mutation in the *Aedes aegypti* voltage-gated sodium channel gene associated with permethrin resistance. Pesti Biochem Physiol 96: 127–131.
- Saavedra-Rodriguez K, Urdaneta-Marquez L, Rajatileka S, Moulton M, Flores AE, et al. (2007) A mutation in the voltage-gated sodium channel gene associated with pyrethroid resistance in Latin American *Aedes aegypti*. Insect Mol Biol 16: 785–798.
- Chang C, Shen W-K, Wang T-T, Lin Y-H, Hsu E-L, et al. (2009) A novel amino acid substitution in a voltage-gated sodium channel is associated with knockdown resistance to permethrin in *Aedes aegypti*. Insect Biochem Mol Biol 39: 272–278.
- Martins AJ, Lima JB, Peixoto AA, Valle D (2009) Frequency of Val1016Ile mutation in the voltage-gated sodium channel gene of *Aedes aegypti* Brazilian populations. Trop Med Int Health14: 1351–1355.

- Martins AJ, Lins RM, Linss, Peixoto AA, Valle D (2009) Voltage-gated sodium channel polymorphism and metabolic resistance in pyrethroid-resistant Aedes aegypti from Brazil. Am J Trop Med Hyg 81: 108–115.
- Rajatileka S, Black WC 4th, Saavedra-Rodriguez K, Trongtokit Y, Apiwathnasorn, et al. (2008) Development and application of a simple colorimetric assay reveals widespread distribution of sodium channel mutations in Thai populations of *Aedes aegypti*. Acta Trop 108: 54–57.
- Lima EP, Paiva MHS, Araújo AP, Silva EVG, Silva UM, et al. (2011) Insecticide resistance in Aedes aegypti populations from Ceará, Brazil. Parasit Vectors, 4: 5.
- Harris AF, Rajatileka S, Ranson H (2010) Pyrethroid resistance in *Aedes aegypti* from Grand Cayman. Am J Trop Med Hyg 83: 277–284.
- Davies TG, Field LM, Usherwood PN, Williamson MS (2007) A comparative study of voltage-gated sodium channels in the Insecta: implications for pyrethroid resistance in Anopheline and other Neopteran species. Insect Mol Biol 16: 361–375.
- Abbot WS (1925) Method of computing the effectiveness of an insecticide. J Econ Entomol 18: 265–267.
- Livak KJ (1984) Organization and mapping of a sequence on the Drosophila melanogaster X and Y chromosomes that is transcribed during spermatogenesis. Genetics 107: 611–634.
- 26. Yanola J, Somboon P, Walton C, Nachaiwieng W, Somwang P, et al. (2011) High-throughput assays for detection of the F1534C mutation in the voltagegated sodium channel gene in permethrin-resistant *Aedes aegypti* and the distribution of this mutation throughout Thailand. Trop Med Int Health 16: 501–509.
- Chang C, Huang X, Chang P, Wu H, Dai SM (2012) Inheritance and stability of sodium channel mutations associated with permethrin knockdown resistance in *Aedes aegypti*. Pest Biochem Physiol, 104: 136–142.
- Tamura K, Peterson D, Peterson N, Stecher G, Nei M, et al. (2011) MEGA5: Molecular Evolutionary Genetics Analysis using Maximum Likelihood, Evolutionary Distance, and Maximum Parsimony Methods. Mol Bio Evo 28: 2731–2739.
- Gaunt TR, Rodríguez S, Day IN (2007) Cubic exact solutions for the estimation of pairwise haplotype frequencies: implications for linkage disequilibrium analyses and a web tool 'CubeX'. BMC Bioinformatics 8: 428.
- Singh RK, Dhiman RC, Mittal PK, Dua VK (2011) Susceptibility status of dengue vectors against various insecticides in Koderma (Jharkhand), India. J Vector Borne Dis 48: 116–118.
- Katyal R, Tewari P, Rahman SJ, Pajni HR, Kumar K, et al. (2001) Susceptibility status of immature and adult stages of *Aedes aegypti* against conventional insecticides in Delhi, India. Dengue Bulletin 25: 84–87
- Mourya DT, Gokhale MD, Chakraborti S, Mahadev PV, Banerjee K (1993) Insecticide susceptibility status of certain populations of *Aedes aegypti* mosquito from rural areas of Maharashtra state. Indian J Med Res. 97: 87–91.
- 33. Shetty V, Sanila D, Shetty NJ (2012) Insecticide susceptibility status in three medically important species of mosquitoes, *Anopheles stephensi*, *Aedes aegypti* and *Culex quinquefasciatus*, from Bruhat Bengaluru Mahanagara Palike, Karnataka, India. Pest Manag Sci 69: 257–267.
- Dev V, Khound K, Tewari GG (2014) Dengue vectors in urban and suburban Assam, India: entomological observations. Southeast J Public Health 3: 51–59.
- Bingham G, Strode C, Tran L, Khoa PT, Jamet HP (2011) Can piperonyl butoxide enhance the efficacy of pyrethroids against pyrethroid-resistant *Aedes* aegypti? Trop Med Int Health 16: 492–500.
- Kangang B, Marcombe S, Chandre F, Nchoutpouen E, Nwane P, et al. (2011) Insecticide susceptibility of *Aedes aegypti* and *Aedes albopictus* in Central Africa. Parasit Vectors 4:79.

- Karunaratne SH, Weeraratne TC, Perera MD, Surendran SN (2013) Insecticide resistance and, efficacy of space spraying and larviciding in the control of dengue vectors *Aedes aegypti* and *Aedes albopictus* in Sri Lanka. Pestic Biochem Physiol 107: 98–105.
- Marcombe S, Mathieu RB, Pocquet N, Riaz MA, Poupardin R, et al. (2012) Insecticide resistance in the dengue vector *Aedes aegypti* from Martinique: distribution, mechanisms and relations with environmental factors. PLoS One 7:e30989.
- Ocampo CB, Salazar-Terreros MJ, Mina NJ, McAllister J, Brogdon W (2011) Insecticide resistance status of *Aedes aegypti* in 10 localities in Colombia. Acta Trop 118: 37–44.
- Yaicharoen R, Kiatfuengfoo R, Chareonviriyaphap T, Rongnoparut P (2005) Characterization of deltamethrin resistance in field populations of *Aedes aegypti* in Thailand. J Vector Ecol 30: 144–150.
- Van den Berg H, Zaim M, Yadav RS, Soares A, Ameneshewa B, et al. (2012) Global trends in the use of insecticides to control vector-borne diseases. Environ Health Perspect. 120: 577–582.
- Ranson H, Burhani J, Lumjuan N, Black WC (2010) Insecticide resistance in dengue vectors. TropIKA.net J 1: 1. (http://journal.tropika.net/pdf/tropika/ vln1/a03vln1.pdf) [accessed: 28 May 2014].
- Kawada H, Higa Y, Komagata O, Kasai S, Tomita T, et al. (2009) Widespread distribution of a newly found point mutation in voltage-gated sodium channel in pyrethroid-resistant *Aedes aegypti* populations in Vietnam. PLoS Negl Trop Dis 3(10): e000527.
- 44. Linss JG, Brito LP, Garcia GA, Araki AS, Bruno RV, et al. (2014) Distribution and dissemination of the Val1016Ile and Phe1534Cys kdr mutations in Aedes aegypti Brazilian natural populations. Parasit Vectors 7: 25.
- Aponte HA, Penilla RP, Dzul-Manzanilla F, Che-Mendoza A, López AD, et al. (2013) The pyrethroid resistance status and mechanisms in *Aedes aegypti* from the Guerrero state, Mexico, Pest Biochem Physiol 107: 226–234.
- 46. Stenhouse SA, Plernsub S, Yanola J, Lumjuan N, Dantrakool A, et al. (2013) Detection of the V1016G mutation in the voltage-gated sodium channel gene of *Aedes aegypti* (Diptera: Culicidae) by allele-specific PCR assay, and its distribution and effect on deltamethrin resistance in Thailand. Parasit Vectors 6: 253.
- Du Y, Nomura Y, Satar G, Hu Z, Nauen R, et al. (2013) Molecular evidence for dual pyrethroid-receptor sites on a mosquito sodium channel. Proc Natl Acad Sci U S A 110: 11785–11790.
- Plernsub S, Stenhouse SA, Tippawangkosol P, Lumjuan N, Yanola J, et al. (2013) Relative developmental and reproductive fitness associated with F1534C homozygous knockdown resistant gene in *Aedes aegypti* from Thailand. Trop Biomed 30: 621–630.
- Srisawat R, Komalamisra N, Eshita Y, Zheng M, Ono K, et al. (2010) Point mutations in domain II of the voltage-gated sodium channel gene in deltamethrin-resistant *Aedes aegypti* (Diptera: Culicidae). Appl Entomol Zool 45: 275–282.
- Singh OP, Bali P, Hemingway J, Subbarao SK, Dash AP, et al. (2009) PCRbased methods for the detection of L1014 kdr mutation in Anopheles culicifacies sensu lato. Malar J 8: 154.
- 51. Bass C, Nikou D, Donnelly MJ, Williamson MS, Ranson H, et al. (2007) Detection of knockdown resistance (kdr) mutations in Anopheles gambiae: a comparison of two new high-throughput assays with existing methods. Malar J 6: 111.